Author’s response to reviews

Title: Predictors of one-year outcomes in chronic heart failure: the portrait of a middle income country

Authors:

Luciana Gioli-Pereira (lu_gioli@yahoo.com)
Fabiana Marcondes-Braga (fgmarcondes@gmail.com)
Sabrina Bernardez-Pereira (bernardezsabrina@gmail.com)
Fernando Bacal (fbacal@uol.com.br)
Fábio Fernandes (fabio.fernandes@incor.usp.br)
Alfredo Mansur (ajmansur@incor.usp.br)
Alexandre Pereira (alexandre.pereira@incor.usp.br)
José Krieger (jose.krieger@incor.usp.br)

Version: 3 Date: 11 Oct 2019

Author’s response to reviews:

About the role of funder, we would like to explain that it was only financial support. The design of the study and collection, analysis, and interpretation of data and in writing the manuscript were realized by the authors.